# WORKING PAPERS 6 | 2012

### THE DYNAMICS OF CAPITAL STRUCTURE DECISIONS

Paula Antão Diana Bonfim

March 2012

The analyses, opinions and findings of these papers represent the views of the authors, they are not necessarily those of the Banco de Portugal or the Eurosystem





Banco de Portugal EUROSYSTEM

#### BANCO DE PORTUGAL

Av. Almirante Reis, 71 1150-012 Lisboa www.bportugal.pt

#### Edition

Economics and Research Department

#### Pre-press and Distribution

Administrative Services Department Documentation, Editing and Museum Division Editing and Publishing Unit

#### Printing

Administrative Services Department Logistics Division

Lisbon, March 2012

#### Number of copies

80

ISBN 978-989-678-120-0 ISSN 0870-0117 (print) ISSN 2182-0422 (online) Legal Deposit no. 3664/83

# The dynamics of capital structure decisions<sup>\*</sup>

Paula Antão

Diana Bonfim

Banco de Portugal and Nova SBE

Banco de Portugal and ISEG-UTL

March 8, 2012

#### Abstract

In this paper we explore the process of convergence to firms' target leverage ratios. Using a unique dataset of micro, small, medium and large firms, we find that this process is very fast, most notably for smaller firms. We further explore these results by analyzing different convergence trajectories. We find that firms that are currently below their target leverage ratio take more time to reach this target than firms with a symmetrical departure point. Furthermore, smaller firms are able to converge faster to their optimal capital structure, regardless of whether they have to increase or decrease their current leverage ratios. Using a duration analysis framework, we also find that firms that have to increase debt to reach their target leverage ratio take more time to do so if they have more free cash-flow.

JEL Codes: G32.

Keywords: Leverage, Trade-off theory, Duration Analysis

<sup>\*</sup>We are thankful to Nuno Alves, Jan Bartholdy, Frederico Belo, Lloyd Blenman, Mário Centeno, Paolo Colla, Ralf Elsas, Ana Cristina Leal, Pedro Portugal, Nuno Ribeiro, Carlos Santos, Frederiek Schoubben and to participants in EFMA 2010, PFN 2010 and WFC 2010 for insightful comments and discussions. The analyses, opinions and findings of this paper represent the views of the authors, which are not necessarily those of Banco de Portugal.

## 1 Introduction

In the corporate finance literature, there is a wide debate on whether firms have an optimal debt structure and, if they do, how do they converge to this target leverage ratio (Fama and French, 2002, Leary and Roberts, 2010, De Angelo et al, 2011). One of the theoretical pillars of this debate is the trade-off model, which argues that firms choose the optimal level of debt by trading off the benefits of debt against its costs. According to this theory, there is an optimal level of debt which is attained when the marginal benefit equals the marginal cost of an additional unit of debt. More recently, this literature has also focused on the speed of adjustment towards the target leverage ratio, examining the process of convergence to this optimal level of debt (Leary and Roberts, 2005, Flannery and Rangan, 2006, Strebulaev, 2007, Lemmon et al, 2008, Huang and Ritter, 2009).

In this paper we contribute to the current debate by examining several aspects of capital structure decisions. First, we explore the determinants of firms' leverage choices, finding a significant negative relation between profitability and leverage, as in Graham (2000), Fama and French (2002) or Korajczyk and Levy (2003). Second, we test one of the main predictions of the trade-off theory, i.e., whether firms converge to an optimal leverage ratio. We observe that firms in the sample do converge to a target leverage ratio and that the speed of adjustment is actually very fast, when compared to previous evidence. Furthermore, we observe that this speed of adjustment is significantly faster for smaller firms. This latter result may help to explain the faster speed of adjustment when compared to other studies, given that our database is mainly comprised of micro, small and medium firms, whereas most of the literature considers mainly large and publicly traded firms. We argue that this fast speed of adjustment could also reflect the use of firm fixed effects, as discussed by Flannery and Rangan (2006) and Lemmon et al (2008).

In addition, we explore in more detail the different possible adjustment trajectories towards the target. Unlike the first two issues addressed, these different adjustment paths have not been thoroughly explored in the capital structure literature (to the best of our knowledge, the only exceptions are Byoun, 2008, and De Jong et al, 2011). We argue that the direction in which the adjustment towards the target is performed may not be irrelevant when estimating the speed of convergence to the target leverage ratio. As such, we consider

that there may be substantial differences in increasing or decreasing leverage to converge to the optimal leverage level. In the first case, the increase in leverage is not entirely a decision of the firm, as it involves finding creditors (either banks or bondholders) willing to provide more credit to the firm. As such, this positive adjustment may be associated with significant costs and frictions. In the second situation, when firms want to decrease the current leverage ratio to reach their optimal target, they are not so dependent on a negotiation with third parties, as the decrease of the amount of debt outstanding is mainly the firms' own decision, even if not necessarily costless. Given these differences in adjustment paths, it is possible that there is some asymmetry in the speed of adjustment, which may be significantly faster when firms want to decrease their leverage. Indeed, we find that firms with a target leverage ratio below their current leverage are able to reach this target faster than firms with a symmetrical departure point. There are several possible explanations for these results. On one hand, this result may reflect the existence of financial constraints in issuing new financial liabilities. On the other hand, firms may prefer to operate (slightly) below their optimal leverage ratio in order to preserve some debt capacity. In addition, for firms operating above their optimal leverage ratio, the faster speed of adjustment associated with a decrease in leverage may reflect a higher flexibility in decreasing outstanding debt. These results are broadly consistent with those of Byoun (2008) and De Jong et al (2011), who also analyze different convergence trajectories. We contribute to extend existing available evidence on this issue by further exploring differences in adjustment theories for firms in different size categories. While most previous studies focus on large publicly traded firms, our dataset allows us to distinguish convergence trajectories of micro, small, medium and large firms. We obtain consistent evidence that the speed of adjustment is decreasing with firm size in all trajectories considered. This shows that smaller firms are able to converge faster to their optimal capital structure, regardless of whether they have to increase or decrease their current leverage ratios.

Finally, we further explore this issue by analyzing the determinants of the time firms take to get close to their target leverage ratio using a duration analysis framework. We find that these determinants do not differ significantly when firms have to increase or decrease their leverage to reach their optimal capital structure, with the exception of the cash flow ratio. In fact, firms that have to increase debt to reach their target take a longer time to do so if they have more free cash flow, given that they face a less severe pressure to obtain external funds. In contrast, firms which aim to reduce debt are able to do so faster if they are able to internally generate more cash flow. This article is organized as follows. Section 2 reviews the literature on capital structure decisions. Section 3 briefly characterizes the dataset used. Section 4 analyses the determinants of firms' leverage ratios and Section 5 discusses whether firms converge to an optimal leverage ratio. Given the results obtained, in Section 6 different adjustment trajectories towards the target are explored in further detail. Finally, Section 7 summarizes our main findings.

# 2 How do firms choose their capital structure?

Since Modigliani and Miller's (1958) irrelevance proposition, firm's capital structure decisions have been intensely investigated. The irrelevance proposition states that under strict assumptions, among which is the absence of corporate taxes, the structure of capital is irrelevant to the determination of a company's value. The assumption on taxes proved to be crucial for the irrelevance proposition. In fact, a few years later, Modigliani and Miller (1963) concluded that the introduction of corporate taxes and the possibility of deducting interest on debt from taxable profits would induce firms to be completely financed by debt. However, as this is not usually observed, several authors, including Modigliani and Miller themselves in Modigliani and Miller (1963), argued that bankruptcy costs, and other costs associated with debt, could explain why firms were not totally financed by debt. This discussion on the benefits and costs of debt is central to the trade-off theory of capital structure. According to this theory, there are forces leading firms to less leverage, for instance bankruptcy costs, and forces leading to more leverage, among them the above mentioned tax benefits of debt and the agency costs of free cash flow. The combination of these forces results in the existence of a target leverage at which the value of firms is maximized.

The main predictions of this theory on leverage ratios are related with the profitability of firms. Profitability should have a positive impact on leverage, as it contributes to a decrease in bankruptcy costs. In addition, more profitable firms benefit more from the tax benefits of debt (DeAngelo and Masulis, 1980). As these firms have more free cash-flow, the existence of debt payments also helps to reduce agency costs of equity, by aligning the interests of managers and shareholders (Jensen and Meckling, 1976, and Jensen, 1986). Besides profitability, there are other characteristics of firms that help to explain target leverages. According to theory, bankruptcy costs are expected to be lower for firms with more tangible assets, as these could be used as collateral. In addition, the existence of depreciation expenses helps to explain less leverage, as these expenses result in tax benefits. Finally, in contrast with the above-mentioned agency models, firms with more investments would have less free cash flow for managers to allocate for their own benefit. Hence, for firms with more investments, debt is not as important as a way to monitor and constrain the actions of managers.

This theory has been tested empirically in several papers that evaluate whether firms converge to a target leverage ratio, focusing on the estimation of the speed of adjustment. There is a significant dispersion in the estimated speed at which firms converge to their optimal capital structure. Fama and French (2002) find a relatively slow speed of adjustment (between 7 and 18 per cent). This slow convergence speed raises questions about whether firms are actually moving towards a target leverage ratio. More recently, Huang and Ritter (2009) obtained similar estimates for the speed of adjustment (around 17 per cent), though using a different estimation methodology.

In turn, Flannery and Rangan (2006) and Lemmon et al (2008) find higher speeds of adjustment. By using a different specification from that used by Fama and French (2002), Flannery and Rangan (2006) obtain estimates close to 34 per cent, i.e., each year the firm is one third closer to its optimal leverage ratio. Furthermore, they conclude that more than half of the changes in firms' capital structure may be attributable to the trade-off theory, while other theories may explain only around 10 per cent of these changes. Using a pooled OLS estimator, Lemmon et al (2008) find a relatively low speed of adjustment (13 to 17 per cent), though this estimate increases significantly when they use a fixed effects estimator (36 to 39 per cent). Both Flannery and Rangan (2006) and Lemmon et al (2008) show that using firm fixed effects leads to significantly higher estimates of the speed of adjustment, when compared to a simple pooled OLS estimation. According to their arguments, neglecting firm fixed effects might lead to biased estimates of the speed of adjustment when firms have relatively stable (and unobserved) variables that contribute to a stable leverage ratio.

There are other arguments in the literature to explain why some studies find relatively low speeds of adjustment. For instance, De Angelo et al (2011) explain the slow speed of adjustment with the hypothesis that firms deviate deliberately from their optimal convergence path to the target to issue transitory debt necessary to finance investments. By excluding this effect, these authors obtain a much higher speed of adjustment (60 per cent, compared to 14 per cent if this effect is not considered). In turn, Strebulaev (2007) argues that firms adjust debt infrequently due to transaction costs, thus often deviating from their optimal leverage ratios. Leary and Roberts (2005) also discuss the role of adjustment costs, arguing that it may not be optimal for firms to continuously rebalance their capital structure. More recently, some discussion emerged on whether the speed of convergence to a target leverage ratio should be different depending on whether firms have to make a positive or a negative adjustment in their current leverage ratio to reach this target. Byoun (2008) finds that firms that have to increase leverage to reach their target record lower speeds of adjustment than firms that have to decrease leverage. This difference is associated with the preference of some firms to preserve debt capacity for future financing needs when the leverage ratio is below the target, thus resulting in a slower adjustment speed. In contrast, firms that are above target and have a financial surplus are able to adjust faster. Similar results are obtained by De Jong et al (2011), who also analyze situations where the trade-off theory conflicts with other capital structure theories.

Some authors have recently analyzed the role of macroeconomic and institutional conditions in the speed of adjustment. Cook and Tang (2010) observe that firms are able to adjust their capital structure faster when macroeconomic conditions are favorable, while Hanousek and Shamshur (2011) argue that firm leverage is very stable even when there are substantial changes in macroeconomic conditions, showing that financial constraints play a key role in explaining the convergence to the target. In turn, Oztekin and Flannery (2012) find that legal and financial features are correlated with the speed of adjustment, as better institutions are associated with lower transaction costs.

It should be noted that, besides the trade-off theory, there are other theories and models that analyze how do firms choose their capital structure. One of the most well-known is the pecking order theory developed by Myers (1984), using the Myers and Majluf (1984) setting of asymmetric information. This theory argues that firms are likely to finance their investments primarily with internal financing, in order to prevent firms from being exposed to the asymmetric information problem. If outside capital is needed, firms are likely to issue debt securities first, that is, those paying a predefined remuneration, which entails lower risk. Only when the firm's debt capacity is reached should the firm consider equity, as it is much riskier and investors would factor in a bigger discount. There is a wide empirical literature that tests several of the predictions of the pecking order theory, contrasting them with the predictions of the trade-off model. Some authors find evidence mainly in favor of the pecking order theory (Shyam-Sunder and Myers, 1999, and Frank and Goyal, 2003), while others show that both theories are relevant and consider complementary dimensions of firms' capital structure decisions (Fama and French, 2002, 2005, Byoun, 2008, Leary and Roberts, 2010, De Jong et al, 2011). In addition, there are two more recent explanations of capital structure decisions, namely the market timing theory by Baker and Wurgler (2002) and the mechanical stock price explanation by Welch (2004). Baker and Wurgler (2002) argue that managers tend to "time the market" by issuing shares when the equity market is perceived as more favorable. The Welch (2004) explanation of capital structure is based on share price fluctuations, arguing that managers simply let market leverage ratios change because of share price fluctuations.

Our paper focuses on testing some of the predictions of the trade-off theory. Our contribution to the existing literature lies on the analysis of different trajectories of adjustment towards the target. Furthermore, we analyze the determinants of the time firms take to reach their target leverage ratio using a duration analysis framework, to better understand why some firms are able to converge faster to their optimal capital structure.

## 3 Data

Most of the empirical research on capital structure relies on data for large publicly traded firms. In this paper, we use a dataset available at Banco de Portugal, the Central Balance Sheet Database, which has a wide coverage for micro, small, medium and large firms. This database provides detailed accounting information on Portuguese firms, and is used mostly for economic and statistical purposes, covering around 390 000 firms. Reporting was not compulsory before 2006. Despite that, the database covers around 60 per cent of total gross value added in the Portuguese economy up to 2005, with larger firms being covered more exhaustively than small and medium-sized ones. Even though this bias constitutes a shortcoming, the database is still an extremely rich and unique dataset on non-financial corporations.

From 2006 onwards the dataset covers all firms operating in Portugal, based on a unified report submitted to the Ministry of Finance, Ministry of Justice, Statistics Portugal and Banco de Portugal. In 2006 firms were asked to report information for the previous fiscal year and, as a result, the information in the Central Balance Sheet database from 2005 onwards refers to all companies operating in Portugal, instead of just to a representative sample. The use of data from the Central Balance Sheet database enables to take into account differences in firm size, economic sector and age.

In Table 1 we present the debt structure of all the firms included in the database. We observe that bank loans are the main source of external finance for the companies included in the sample, accounting for more than 55 per cent of total debt. Trade credit accounts for

slightly more than one fifth of firms' debt, though its importance has declined during the sample period. Debt securities represent a smaller fraction of firms' debt (less than 10 per cent), even for the larger firms in the sample, reflecting the fact that very few Portuguese firms raise funding in wholesale debt markets.

In order to enhance the quality of the data used for our analysis and to obtain nonspurious regression results, we need to apply some filters to the data. First, we remove from the dataset observations with a negative value of assets and observations with zero employees. We also remove observations for which there are less than two consecutive years of data and with no information on firm foundation date. Moreover, to deal with spurious outlier observations, we winsorize observations below (and above) the 1st (99th) percentile for some relevant variables. We end up with a total number of about 350 000 observations for the period from 1990 to 2007. These observations correspond to about 48 000 firms. On average, we observe firms for 9 years.

Table 1 also displays summary statistics for the leverage ratio, defined as bank loans and bonds as a percentage of total assets. When the whole sample is considered, the leverage ratio is, on average, 32 per cent, having remained relatively stable during the sample period. This stability is in line with evidence found by Lemmon et al (2008). When only the reduced sample is considered, after applying the abovementioned filters, the leverage ratio decreases to around 20 per cent. Moreover, the median values for this sub-sample are much lower, standing at 4 per cent.

We created four classes of firms with different sizes considering simultaneously the value of sales and the number of employees (firm size definitions are presented in Table 2). Most of the firms in the sample are micro firms, having less than 10 employees and less than 2 million euros in turnover. As it would be expected, most of these firms do not use external finance, more specifically bonds and bank loans, as the median leverage ratio for these firms is zero during the sample period. Small firms also represent a significant part of the sample. Their median leverage ratio stands at 8 per cent, referring almost exclusively to bank loans. Medium-sized firms are the most leveraged (their median leverage ratio is 14 per cent). Finally, large firms show a slightly lower median leverage ratio (12 per cent). Most bonds are issued by this last group of firms.

We also grouped firms according to their age. The average age of a firm in this dataset is 18 years and 10 per cent of the observations correspond to firms with less than 5 years. On the other hand, the 90th percentile corresponds to firms with 36 years. We defined four age classes according to the percentiles 25, 50 and 75 (see Table 3). We observe that leverage seems to be (non-linearly) increasing with firm age.

Finally, we also examine differences between economic sectors (Table 4), observing that the most leveraged sectors (taking into account median values) are real estate firms (18.4 per cent), followed by utilities (9.9), mining firms (7.5) and construction (6.1).

# 4 Explaining leverage ratios

One preliminary step in this paper is to analyze the determinants of the leverage ratio. Our empirical research strategy is to estimate a fixed effects panel data model such that:

$$\frac{D}{A_{it}} = f_i + \beta_1 + \beta_2 X_{it} + \psi_t + \varepsilon_{it}.$$
(1)

The dependent variable is  $\frac{D}{A_{it}}$ , the leverage ratio, defined as bonds and loans as a percentage of total assets.<sup>1</sup>  $X_{it}$  is a set of relevant firm characteristics which may affect firms' leverage, including  $\frac{Cash-flow}{Assets}$ , Sales Growth,  $\frac{Tangible Assets}{Assets}$ , Assets, Group Dummy, Liquidity,  $R\&D \ Dummy$  and Depreciation. All these variables are firm-specific and time-varying. We control for firm and time fixed effects ( $f_i$  and  $\psi_t$ , respectively).

 $\frac{Cash-flow}{Assets}$  is computed as net earnings before provisions and depreciation, scaled by firms' assets.<sup>2</sup> Sales growth is the year-on-year change of sales, which is included in the regressions to control for firm's growth.  $\frac{Tangible Assets}{Assets}$ , the share of tangible assets in total assets, controls for the asset structure of the firm, and also for the collateral potentially available for debt contracts. Firms whose assets are mostly comprised of intangibles may find it harder to obtain bank financing, thus displaying lower leverage ratios. In fact, as bankruptcy costs play a prominent role in the trade-off theory, asset tangibility is predicted to have a positive impact on leverage. We also consider a dummy variable which considers whether the firm belongs to a group, as this may yield important differences in terms of capital structure decisions, given the possibility of access to intra-group funding. If a firm records assets or liabilities within a group, then this variable takes the value one. In our regressions we also control for *Liquidity*, defined as cash and short-term securities as a percentage of short-term

<sup>&</sup>lt;sup>1</sup>The leverage measure is based on book values. In the capital structure literature, many papers use the market value instead, which is available for publicly traded firms. However, several authors show that using book or market leverage does not yield substantial differences in the results (Fama and French, 2002, Flannery and Rangan, 2006, Huang and Ritter, 2009, De Jong et al, 2011).

<sup>&</sup>lt;sup>2</sup>Alternatively, it would be possible to use profitability measures, such as net earnings over assets. The results obtained with this alternative variable are similar to those associated with the cash-flow ratio.

debt. Another potentially relevant explanatory variable is the R&D Dummy, which takes the value one whenever the firm records some R&D investment. In addition, this variable, together with the variable Depreciation, also helps to control for non-debt tax shields. Thus, we also control for depreciations and provisions, measured as a percentage of total assets. Finally, given the apparent importance of firm size on leverage ratios, we use the logarithm of assets as a control variable as well.

In Table 5 we present a brief statistical description of the variables considered in this analysis and in Table 6 we include a correlation matrix of the same variables.

Table 7 presents our first regression results. In the first column we estimate equation (1). We control, as in all other regressions, for time and firm fixed effects. The coefficients obtained for the control variables are all statistically significant at 5 per cent (except for the *Group Dummy*).

We obtain a significant negative coefficient for cash-flow, thus showing that firms with more available funds will use less external funding than other companies. Many authors also found a significant negative relation between these two variables (see, for example, Graham, 2000, Fama and French, 2002, Korajczyk and Levy, 2003, Frank and Goyal, 2003), including for small and medium enterprises (Sogorb-Mira, 2005, Degryse et al, 2010). In contrast, Korteweg (2010) recently found that more profitable firms have higher optimal debt ratios.

Firms with stronger sales growth show lower leverage ratios, even though this effect is very small. If this variable is seen as a proxy for growth opportunities, this negative coefficient is consistent with the trade-off theory, as risk tends to be higher for these firms, pushing up bankruptcy costs. Firms with more tangible assets (and hence with more collateral potentially available for credit) are also more indebted than other firms, as the trade-off theory predicts. Firm size seems to be extremely important in explaining leverage ratios, as larger firms show much higher leverage ratios than other firms, other firm characteristics being controlled for. This is consistent with the view that larger firms tend to be more diversified and, hence, less volatile, as discussed by Fama and French (2002). Firms with stronger liquidity buffers are also less indebted. In contrast, we observe that firms engaging in R&D activities show higher leverage ratios than others. Finally, firms with more significant depreciations and provisions, as a percentage of their assets, also record higher leverage ratios. This effect does not comply with the predictions of the trade-off theory, as the tax benefits of debt should be less relevant for firms that already have significant depreciation and provisioning expenses. Nevertheless, the results for this specification may be seriously affected by simultaneity issues. In fact, it is possible that there are some unobserved time-varying variables which simultaneously affect the leverage ratio and other firm-specific variables, thus leading to potential endogeneity problems. In order to minimize this problem, we consider an alternative specification, in which all explanatory variables are lagged by one year, such that:

$$\frac{D}{A_{it}} = f_i + \beta_1 + \beta_2 X_{it-1} + \psi_t + \varepsilon_{it}$$
(2)

This specification is presented in the second column of Table 7. The estimated coefficient for cash-flow remains consistent with that previously observed: firms with more available funds are less indebted than other firms, controlling for other relevant firm characteristics. As regards the other firm control variables, there are some differences worth noticing. In particular, *Sales Growth* and *Depreciation* are no longer statistically significant at a 5 per cent level. For all other control variables, the results are generally consistent with those observed in the previous specification.

Our results are broadly in line with those obtained by Fama and French (2002). These authors estimate a model similar to (2) without considering firm-level fixed effects. However, we consider that the inclusion of firm level-fixed effects is crucial, as they control for time invariant unobserved heterogeneity at the firm level.

The results presented in the previous section suggest that the determinants of firm leverage may be considerably different depending on firms' size and age. In order to better explore these possible differences, we estimate the regression with all explanatory variables lagged by one year for different size and age cohorts. The results of these estimations are displayed in Table 8. First, we observe that the estimated coefficient for  $\frac{CF}{A}_{it-1}$  remains negative and statistically significant regardless of firm size (columns 1-4). Moreover, we observe that this coefficient becomes larger, in absolute value, as firm size increases, thus suggesting that large firms with more internal funds available use less external funding than comparable smaller firms. We obtain a similar result when we estimate the regressions by firm age: older firms have a similar behavior to that of larger firms (columns 5-8).

In what concerns the other control variables, the results are broadly consistent with those previously obtained. *Sales Growth*, *Group Dummy* and *Depreciation* are not significant in most firm groups and the remaining variables hold the same signals, when significant.

For robustness purposes, we also estimate the regression for different sectors. In Table 8 we present the results for manufacturing firms, as these represent a large part of the sample (column 9). The results are broadly consistent with those previously obtained and there is a slight improvement in the model's adjustment quality.

Micro firms might be different in some aspects, possibly not being so able to optimize their leverage choices, as they usually face more restrictions in access to funding. To be sure that these firms do not bias the results, we estimate the regression excluding these micro firms, obtaining consistent results (column 10).

It is important to notice that more than 40 percent of the firms in the sample do not rely either on bank or market financing, thus having null leverage ratios. Given the possibility that this feature may affect the results of the estimations, we also present in Table 8 the regression estimated only for firms with positive leverage (column 11). The only difference in the determinants of leverage ratios for this specific group of firms is that *Depreciation* now has a positive significant impact on leverage. Nevertheless, the results of the other control variables remain unchanged. Considering that the decision on whether to seek external funding or not can be made before the choice of the leverage ratio, we also estimate a discrete choice regression to empirically analyze this preliminary decision to use external funds. In this regression, the dependent variable is a binary variable which takes the value one when the firm has positive leverage. The results are also shown in Table 8 (column 12). Firms with positive leverage ratios have lower cash-flow ratios than firms with no external funding, other characteristics controlled for. All other firm characteristics considered yield results consistent with those previously obtained.

Furthermore, we also test alternative definitions of leverage, considering only long-term debt (bonds and bank loans), as done by Flannery and Rangan (2006), and a broader definition of leverage, that also includes debt from suppliers of tangible assets (columns 13 and 14, respectively). Again, the results remain broadly consistent with those obtained in the other specifications.

Finally, we also control for the total amount of debt a firm has outstanding on the group it belongs to, instead of having a simple dummy variable indicating whether or not the firm has debt from the group (column 15). The impact of this variable on leverage is now statistically significant, though not economically meaningful.

All in all, our results are broadly consistent with those obtained in the literature on the determinants of leverage ratios. We find that leverage ratios are higher for firms with lower profits, more tangible assets, lower liquidity ratios and positive R&D investments, as well as for larger firms.

# 5 Do firms have a target leverage ratio?

As discussed in Section 2, firms may have target leverage ratios to which they converge over time. This is a central result of the trade-off theory. In order to test whether this conclusion is valid for the firms in our dataset, we estimate a two-step regression, in a spirit similar to that of Fama and French (2002). In the first step, we estimate a regression as defined in equation (2). However, given that the distribution of the leverage ratio has a clear discontinuity, with more than 40 percent of the firms having null leverage ratios, we have to treat differently the firms without leverage. To control for that effect, we include a dummy variable (contemporaneous and lagged) that takes the value one only when the firm has positive leverage. We then compute the target leverage ratio for all firms, defined as the fitted values of this regression. When firms have a negative target leverage ratio, we consider that their target is zero.

In the second-step regression, we use the fitted values of the first-step as a proxy for the target leverage (TL) in a partial adjustment model. In this model, changes in leverage ratios partly reflect the difference between firms' target leverage and the previous year's observed leverage ratio. In this second step we estimate the following regression:

$$\Delta \frac{D}{A_{it}} = f_i + \beta_1 + \beta_2 a djust_{it} + \beta_3 \Delta \frac{CF}{A_{it}} + \beta_4 \Delta \frac{CF}{A_{it-1}} + \beta_5 \frac{Inv}{A_{it}} + \beta_6 \frac{Inv}{A_{it-1}} + \psi_t + \varepsilon_{it} \quad (3)$$

where,

$$adjust_{it} = \left[TL_{it} - \frac{D}{A_{it-1}}\right]$$

 $\frac{CF}{A}$  is the ratio of cash-flow to assets, and the variable  $\frac{Inv}{A}$  measures investment expenditures scaled by total assets. The estimation of this partial adjustment model allows us to test some of the predictions of the trade-off theory given that, according to this theory, firms have target leverage ratios and move toward the target over time.<sup>3</sup> Hence,  $\beta_2$ , which measures the speed of adjustment, should be positive. The investment and cash flow variables are included to control for any short-term movements in leverage away from the target, in line with the work of Fama and French (2002). The results of this estimation are presented in

<sup>&</sup>lt;sup>3</sup>In the presence of a cointegration relationship, a different estimation approach should be followed. However, as the results presented in Table 9 use almost 40,000 firms, observed over 5 years, on average, the stationarity and cointegration tests for panel datasets cannot benefit from the necessary asymptotic properties. In this case, as the panel dataset has a small T and a large N, the existing panel data procedures are sufficient to consider very general temporal correlation patterns (see Hsiao, 2003).

the first column of Table 9. The adjustment variable has a coefficient of 0.62, which means that every year firms get roughly 60 percent closer to their target leverage ratio. Hence, the results are clearly in favor of an adjustment toward the target, thus providing evidence supporting the trade-off theory. Our results are consistent with the ones in Fama and French (2002), Flannery and Rangan (2006), Antoniou et al (2008) and Huang and Ritter (2009), although the speed of adjustment found by these authors is much slower than ours. Flannery and Rangan (2006) and Lemmon et al (2008) found that using firm fixed effects leads to significantly higher estimates of the speed of adjustment, when compared to a simple pooled OLS estimation.<sup>4</sup> In both papers, the authors argue that it is appropriate to account for unobserved firm heterogeneity through fixed effects when firms have relatively stable (and unobserved) variables that contribute to a stable leverage ratio. Neglecting these fixed effects might lead to biased estimates of the speed of adjustment.

Given that firm size may be associated with different adjustment capabilities, we also run the second step regression for different firm size groups (columns 2-5). Interestingly, we observe that there are indeed different adjustment speeds (in contrast with what is found by Flannery and Rangan, 2006, who do not find significant differences across firm size using a large sample of publicly traded firms). In fact, we find that smaller firms are able to adjust much faster towards their target leverage ratio. The adjustment variable has a coefficient of 0.73 for micro firms and of only 0.45 for large firms.<sup>5</sup> This difference between smaller and larger firms may also help to explain the differences between our results and those obtained by Fama and French (2002), given that their dataset covers only large firms.

In columns 6 to 9 we run the same regressions for firms in different age quartiles. As observed for firm size, the speed of adjustment is decreasing with firm age: younger firms converge much faster to their target leverage ratios. Indeed, for firms in the first quartile this adjustment is almost immediate, though it should be noted that these firms display very low leverage ratios, as shown in Table 3. These results show that younger and smaller firms are able to reach their target leverage ratios faster, possibly reflecting more flexibility in the adjustment of their capital structure.

For robustness purposes, we estimate these regressions for other sub-groups of firms and for other leverage definitions. In column 10 we present the results only for the most representative sector in the sample, manufacturing firms, obtaining an adjustment speed

<sup>&</sup>lt;sup>4</sup>The same pattern is observed in our sample. The results are available upon request.

 $<sup>^{5}</sup>$ We conducted the same exercise but estimating target leverage ratios in separate regressions according to firm size. The results remain consistent.

of 0.59.<sup>6</sup> As micro firms may face more active financing constraints that compromise their ability to optimize their leverage choices (i.e., borrowing close to the limit granted by banks), we also exclude them from the regressions, to ensure that these firms do not bias the results. Again, the results are consistent (column 11). Given the discontinuity in the distribution of leverage ratios, we estimate the same regression including a dummy for firms with positive leverage (columns 12 and 13, respectively). The results obtained are perfectly consistent with the ones for the full sample. Finally, as in the previous section, we also test alternative definitions of leverage. The speed of adjustment is faster when only long term debt is considered (column 14). This result is in line with DeAngelo et al (2011), who find that firms may deviate from their adjustment paths by issuing transitory debt to finance investments. When these transitory financing decisions are excluded, DeAngelo et al (2011) find that the speed of adjustment is significantly faster. In turn, for the broader definition of leverage (that includes also debt from suppliers of tangible assets), the results are globally in line with those obtained with the baseline definition of leverage (column 15).

## 6 Exploring the adjustment towards the target

The highly significant speed of adjustment found in the previous section may not be the same for all firms, depending on how are they currently positioned vis-à-vis their optimal leverage ratio. For instance, if a firm has to increase debt to reach its target, it will have to find new creditors (either banks or bondholders) willing to provide more credit to the firm. In contrast, if the firm needs to decrease its leverage ratio to reach an optimal capital structure (either by letting loans reach their maturity or by doing pre-payments on loans outstanding), it will not be so dependent on the negotiation with third parties, as the decrease of the amount of debt outstanding is mainly the firms' own decision. As such, the positive adjustment (i.e., when firms have to increase leverage) may be associated with higher costs and frictions than a negative adjustment. Furthermore, some firms may prefer to operate (slightly) below their optimal leverage ratio in order to preserve some debt capacity. Hence, given these differences in adjustment paths, we argue that there may be some asymmetry in the speed of adjustment, which may be significantly faster when firms want to decrease their leverage.

<sup>&</sup>lt;sup>6</sup>In addition to this, we also estimate separately target leverage ratios for all the sectors considered in the sample. The results are broadly consistent and are available upon request.

This issue had not, until recently, been explored in the empirical capital structure literature. To the best of our knowledge, the only exceptions are Byoun (2008) and De Jong et al (2011), who analyze firms' adjustment trajectories by considering simultaneously several dimensions and predictions of the trade-off and pecking order theories. Byoun (2008) finds that firms that have to increase leverage to reach their target record slower speeds of adjustment than firms that have to decrease leverage. This difference is associated with the preference of some firms to preserve debt capacity for future financing needs when the leverage ratio is below the target, thus resulting in a slower adjustment speed. In contrast, firms that are above target and have a financial surplus are able to adjust faster. Similar results are obtained by De Jong et al (2011), who also analyze situations where the trade-off model conflicts with the pecking order theory.

In this section we shed some more light on this issue, by focusing exclusively on the estimation of the speed of adjustment for firms with different initial positions in relation to their target leverage ratio. In addition, given the differences in adjustment speeds found for firms across size categories, we evaluate the speed of adjustment in separate trajectories for these different groups of firms. Finally, to better understand why some firms converge faster than others to their target leverage ratio, we explore the determinants of this process using a duration analysis framework.

### 6.1 Positive and negative adjustment paths

As discussed above, positive and negative adjustment paths may be associated with some asymmetry in the speed of adjustment. In this section we analyze whether the speed of adjustment is significantly faster when firms want to decrease their leverage.

In Table 10 we provide some evidence on these adjustment paths. Even though our main goal is to distinguish firms with positive and negative adjustments, we have to consider two different types of positive adjustments. More specifically, we distinguish firms that have to make a positive adjustment to reach their target having no leverage as a departure point, from those firms that already have debt and have to increase their current leverage ratio. We consider that this distinction is relevant because firms that are currently not borrowing may face more difficulties in reaching their optimal leverage ratio (or, alternatively, they may prefer to remain debt-free, even if that is not consistent with the predicted optimal capital structure for these firms). Firms that do not make any adjustments from one period to another are excluded from this analysis. We observe that two thirds of the firms are currently below the target leverage ratio, what may suggest that many firms face funding constraints. As expected, the current leverage ratio is much higher for the firms that have to decrease leverage, even though the target leverage ratio for firms with leverage is very similar, regardless of whether they are above or below their target. For firms without leverage, their predicted target is, nevertheless, relatively small (4 per cent, compared to 19 per cent for the remaining firms). In sum, there are fewer firms that need to reduce their current debt levels, even though the magnitude of the adjustment for these firms is larger.

Larger firms display higher leverage ratios, as well as higher targets. There are proportionally more large firms that need to increase their leverage and, furthermore, the magnitude of this adjustment is more significant. Many micro firms do not have debt in their balance sheets (nearly 40 per cent) and their target is comparatively low (2.5 per cent). However, there are also many micro firms that may be considered as over-indebted (33 per cent), having to decrease their observed leverage ratio more than other firms.

In Table 10 we also analyze differences in sectors, observing that there is a significant heterogeneity in leverage ratios and adjustment paths. Whereas in some sectors many firms with debt operate below their target leverage ratios (most notably in mining, manufacturing and education), in others, a large number of firms may be considered as over-indebted (especially real estate, agriculture and construction<sup>7</sup>). The magnitude of adjustment is, by far, larger for real estate firms.

The next step is to explore these differences in adjustment trajectories in a multivariate setting. In order to do that, we estimate equation (3) for the three possible adjustment paths: negative adjustment, positive adjustment for firms with leverage and positive adjustment for firms without leverage (Table 11). Comparing firms that currently show positive leverage (columns 1 and 2), we observe that the speed of adjustment is faster for firms that have to decrease leverage to reach their optimal capital structure (63 per cent, against 53 per cent for firms that need to make a positive adjustment). As discussed above, in the absence of financial constraints and other frictions, it would be expected that the speed associated with negative and positive adjustments would be relatively similar. However, as the decision to increase debt is not strictly the firm's own, the speed of positive adjustment should be slower, as it requires finding a borrower willing to finance the firm's investment projects.

<sup>&</sup>lt;sup>7</sup>In the utilities sector, many firms also display leverage ratios above their target. However, some firms in this sector are regulated and capital structure decisions may be affected by regulatory restrictions.

When the three possible adjustment paths are compared, the faster adjustment is associated with firms that currently do not have leverage (73 per cent). This result is somewhat unexpected, as these firms have to move from a debt-free capital structure to a positive leverage ratio. In other words, these firms need more than to increase the amount of debt outstanding, as they have to engage into a new lending relationship, being more exposed to asymmetric information problems. However, we observe, in Table 10, that this group of firms typically has a very low target leverage ratio, what should explain this convergence speed, as the amounts involved should be relatively small.

In Table 12 we further explore the differences in the speed of adjustment for different trajectories. In Panel A we compare the speed of adjustment for different firm size categories. For all the adjustment paths, the speed is decreasing with firm size, confirming that smaller firms are able to converge faster to their optimal capital structure, regardless of whether they have to increase or decrease their current leverage ratios. The faster adjustment is observed when firms still do not have debt for all firm sizes, except for the larger firms in the sample. However, it should be noted that very few large firms have null leverage ratios.

When we consider the entire distribution of adjustments, our results reflect both very small and very large adjustments. However, some of these small adjustments may be transitory and/or driven by reasons other than the optimization of the capital structure. Furthermore, the economic and financial costs of adjustments of different magnitudes are substantially different, thus making it relevant to explore possible differences in the speed of adjustment. Hence, in Panel B we explore this question, focusing only on firms that have to make significant adjustments to reach their optimal capital structure. To do that, we consider only adjustments above the median. When all firm size categories are considered, there are some interesting changes in the results. The speed of adjustment for firms that have a leverage ratio significantly above the median is now much faster. For these firms, the costs of being significantly over-indebted may be so high that they commit to reducing their debt ratios as fast as possible. In turn, the speed of adjustment for firms without leverage is now substantially lower. This result reinforces our previous argument to explain the fast adjustment process for these firms (based on the fact that these were mainly small adjustments). When only large adjustments are considered, their speed is considerably slower and closer to that of other firms that have to increase leverage ratios. Nevertheless, this speed continues to be fast, if we consider that these firms have to create a lending relationship. Still, these are firms with very low target leverage ratios. When we run these regressions

for firms in different size categories, we continue to observe that, in general, smaller firms adjust faster than larger firms.

In sum, we observe that firms which have a negative adjustment are able to reach their target faster than firms that have to increase leverage. This result seems to suggest the existence of financial constraints in issuing new financial liabilities, as well as relatively high flexibility in decreasing leverage if firms are operating above their equilibrium leverage ratio.

### 6.2 The determinants of the speed of adjustment in a duration

### analysis framework

The results obtained so far show that firms converge rapidly to their target leverage ratio. The speed of adjustment is especially fast for the smallest firms in the sample, what can help to explain the differences between our results and those usually found in the literature, as most available databases include mainly large and publicly traded firms. To further explore these results, we analyze the determinants of the time firms take to get close to their target leverage ratio using a duration analysis framework. As a baseline definition we consider that the firm is close to the target when the observed leverage ratio is within the interval [90%, 110%] of the initial estimated target.<sup>8</sup>

This econometric framework allows us to estimate semiparametric Cox regressions in order to evaluate which firm characteristics are relevant to explain differences in the time firms take to reach their target ratio. The Cox proportional hazard model may be defined as:  $h(t, X(t), Z(t)) = \kappa(X(t), Z(t))h_0(t)$ , where  $\kappa(.)$  is a non-negative function of X(t) and Z(t), and  $h_0(t)$  is defined as the baseline hazard, which is common to all firms (individual hazard functions differ from each other proportionally, as a function of  $\kappa(X(t), Z(t))$ . This is a partly non-parametric approach, given that we can estimate unknown parameters of  $\kappa(.)$  without specifying the form of the baseline hazard. In these regressions, an estimated coefficient lower than 1 should be interpreted as contributing a longer time until the firm reaches its target.

We find that firms which belong to a group or that have higher liquidity ratios take, on average, a longer time to converge to their optimal debt ratio (Table 13, column 1). In turn,

<sup>&</sup>lt;sup>8</sup>For robustness purposes we also consider the broader interval [80%, 120%]. Our approach differs from the one by Leary and Roberts (2005) who also employ duration analysis to assess how do firms rebalance their leverage to stay within an optimal range.

larger firms<sup>9</sup>, as well as those with more tangible assets, more expenses in depreciations and provisions and with some R&D investments are able to reach their targets sooner.

As discussed previously, the rebalancing pattern of firms with positive or negative adjustments may be very different, even though this issue has not been much discussed in the literature. By running separate regressions for the three possible adjustment paths (columns 2-4), we observe that the signal of the determinants does not differ significantly with the exception of the cash flow ratio. In fact, firms that have to increase debt to reach their target take a longer time to do so if they have more available cash flow, given that they face a less severe pressure to obtain external funds. In contrast, firms which aim to reduce debt are able to do so faster if they are capable of internally generating more cash flow, as expected. Sales growth is only significant for firms that have to decrease leverage, thus providing further support to this argument.

Belonging to a group only influences the time to reach the target for the firms that have to increase leverage, starting from a zero leverage ratio. If these firms belong to a group, they take more time to reach this target, what should reflect the fact that these firms are able to have access to funding within the group, thus not feeling so pressured to obtain external funding.

# 7 Concluding remarks

In this paper we examine several dimensions of one of firms' most important financial decisions: capital structure. In order to do so, we explore the information contained in the Portuguese Central Balance Sheet database. We observe that banks are the most important source of long-term debt for Portuguese non-financial corporations, as access to market funding is to some extent limited to larger firms.

Our analysis was performed using two complementary steps. First, we explored the determinants of firms' leverage ratios, finding a significant negative relation between profitability and leverage. Second, we focused our analysis on the speed of adjustment towards an optimal target ratio. The optimal leverage ratio was estimated using the analysis of the

<sup>&</sup>lt;sup>9</sup>It should be noted that the result regarding firm size is not directly comparable to the different speeds of adjustment found in the previous subsection for different firm size categories, as here we are relying on a continuous measure of size, based solely on total assets (whereas the definition of firm size categories is based on turnover and number of employees). When we run this Cox regression for the four size groups, we find that micro and small firms with more total assets converge faster to their target leverage ratio, whereas medium and large firms with more assets show the opposite pattern (results available upon request).

determinants of firms' leverage ratios, in a spirit similar to that of Fama and French (2002). We observe that most firms converge to a target leverage ratio and that the speed of adjustment is actually very fast, when compared to previous evidence. Furthermore, we observe that this speed of adjustment is much faster for smaller firms. This latter result may help to explain the faster speed of adjustment we obtain when compared to other studies, given that our database is mainly comprised of micro, small and medium firms, whereas most of the literature considers mainly large and publicly traded firms.

Given the speed of adjustment found in this and other studies, we further explored the possibility that different adjustment paths are associated with difference convergence speeds. This issue has not been explored in great detail in the capital structure literature. However, we argue that this is an important dimension of the convergence process towards the target, as the adjustments performed may be very different depending on whether the firm is currently above or below its optimal leverage ratio. When we do not distinguish between these two possibilities, only the average behavior is captured. Hence, we consider separately two possible adjustment trajectories, distinguishing between negative and positive adjustments. Furthermore, there may be two different types of positive adjustments, depending on whether or not the firm currently has debt outstanding. We consider that this distinction is relevant because firms that are currently not borrowing may face higher difficulties in reaching their optimal leverage ratio (or may prefer to remain debt-free, even if that is not consistent with the predicted optimal capital structure for these firms).

In our sample, we find a significant asymmetry in positive and negative adjustment trajectories, observing that firms with a target leverage ratio below their current leverage are able to reach this target faster than firms that are currently above this optimal level. There are several possible explanations for these results. On one hand, this result may reflect the existence of financial constraints in issuing new financial liabilities. On the other hand, firms may prefer to operate (slightly) below their optimal leverage ratio in order to preserve some debt capacity. In addition, for firms operating above their optimal leverage ratio, the faster speed of adjustment associated with a decrease in leverage may reflect a higher flexibility in decreasing outstanding debt.

In addition, we further explore the speed of adjustment by analyzing the determinants of the time firms take to get close to their target leverage ratio using a duration analysis framework. We find that the main difference between positive and negative adjustment paths is related to the cash flow ratio. When firms have to increase debt to reach their target, they take a longer time to do so if they have more available cash flow, possibly reflecting the fact that they have less pressure to obtain external funds. In contrast, firms which aim to reduce debt are able to do so faster if they are able to internally generate more cash flow.

The results obtained illustrate the relevance of considering the heterogeneity of firms' characteristics when analyzing capital structure decisions. As most of the empirical research on this topic focuses on large publicly traded firms, several important aspects may be left unexplained when smaller firms are not part of the sample. Furthermore, examining differences in firms' adjustment trajectories towards the optimal capital ratio may also provide important insights, as illustrated by our results.

# References

Antoniou, A., Guney, Y. and Paudyal, K. (2008), The Determinants of Capital Structure: Capital Market-Oriented versus Bank-Oriented Institutions, *Journal of Financial and Quantitative Analysis*, 43, pp. 59-92.

Baker, M. and Wurgler, J. (2002), Market Timing and Capital Structure, *Journal of Finance*, 57, pp. 1-32.

Byoun, S. (2008), How and When Do Firms Adjust Their Capital Structures toward Targets?, *Journal of Finance*, 63, pp. 3069-3096.

Cook, D. and Tang, T. (2010), Macroeconomic Conditions and Capital Structure Adjustment Speed, *Journal of Corporate Finance*, 16, pp. 73–87.

DeAngelo, H. and Masulis, R. (1980), Optimal Capital Structure under Corporate and Personal Taxation, *Journal of Financial Economics*, 57, pp. 3-29.

DeAngelo, H., DeAngelo, L. and Whited, T. (2011), Capital Structure Dynamics and Transitory Debt, *Journal of Financial Economics*, 99, pp. 235-261.

De Jong, A., Verbeek, M. and Verwijmeren, P. (2011), Firms' Debt–equity Decisions When the Static Tradeoff Theory and the Pecking Order Theory Disagree, *Journal of Banking and Finance*, 35, pp. 1303–1314.

Degryse, H., de Goeij, P. and Kappert, P. (2010), The Impact of Firm and Industry Characteristics on Small Firms' Capital Structure, *Small Business Economics, forthcoming.* 

Fama, E. and French, K. (2002), Testing Trade-off and Pecking Order Predictions About Dividends and Debt, Review of Financial Studies, 15, pp. 1-33.

Fama, E. and French, K. (2005), Financing Decisions: Who Issues Stock?, Journal of Financial Economics, 76, pp. 549–582.

Flannery, M. and Rangan, K. (2006), Partial Adjustment Toward Target Capital Structures, *Journal of Financial Economics*, 79, pp. 469-506.

Frank, M. and Goyal, V. (2003), Testing the Pecking Order Theory of Capital Structure, Journal of Financial Economics, 67, pp. 217-248.

Graham, J. (2000), How Big Are the Tax Benefits of Debt?, *Journal of Finance*, 55, pp. 1901-1941.

Hanousek, J. and Shamshur, A. (2011), A Stubborn Persistence: Is the Stability of Leverage Ratios Determined by the Stability of the Economy?, *Journal of Corporate Finance*, 17, pp. 1360–1376.

Hsiao, C. (2003), Analysis of Panel Data, Econometric Society Monographs, Cambridge University Press.

Huang, R. and Ritter, J. (2009), Testing Theories of Capital Structure and Estimating the Speed of Adjustment, *Journal of Financial and Quantitative Analysis*, 44, pp.237-271.

Jensen, M. C. and Meckling, W. H. (1976), Theory of the firm: Managerial behavior, agency costs, and ownership structure, *Journal of Financial Economics*, 3, pp. 305-360.

Jensen, M. C. (1986), Agency Cost Of Free Cash Flow, Corporate Finance, and Takeovers, *American Economic Review*, 76, pp. 323-329.

Korajczyk, R. and Levy, A. (2003), Capital Structure Choice: Macroeconomic Conditions and Financial Constraints, *Journal of Financial Economics*, 68, pp. 75-109.

Korteweg, A. (2010), The Net Benefits to Leverage, *Journal of Finance*, 65, pp. 2575-2619.

Leary, M. and Roberts, M. (2005), Do Firms Rebalance Their Capital Structures?, *Jour*nal of Finance, 60, pp. 2137-2170.

Leary, M. and Roberts, M. (2010), The Pecking Order, Debt Capacity, and Information Asymmetry, *Journal of Financial Economics*, 95, pp. 332–355.

Lemmon, M., Roberts, M. and Zender, J. (2008), Back to the Beginning: Persistence and the Cross-Section of Corporate Capital Structure, *Journal of Finance*, 63, pp. 1575-1608.

Modigliani, F. and Miller, M. (1958), The Cost of Capital, Corporation Finance and the Theory of Investment, *American Economic Review*, 48, pp. 261-97.

Modigliani, F. and Miller, M. (1963), Corporate Income Taxes and the Cost of Capital: a Correction, *American Economic Review*, 53, pp. 433–443.

Myers, S. (1984), The Capital Structure Puzzle, Journal of Finance, 39, pp. 575-592.

Myers, S. and Majluf, N. (1984), Corporate Financing and Investment Decisions When Firms Have Information that Investors do Not Have, *Journal of Financial Economics*, 47, pp. 187-221.

Oztekin, O. and Flannery, M. (2012), Institutional Determinants of Capital Structure Adjustment Speeds, *Journal of Financial Economics*, 103, pp. 88-112.

Shyam-Sunder, L. and Myers, S. (1999), Testing static tradeoff against pecking order models of capital structure, *Journal of Financial Economics*, 51, pp. 219-244.

Sogorb-Mira, F. (2005), How SME Uniqueness Affects Capital Structure: Evidence From A 1994–1998 Spanish Data Panel, *Small Business Economics*, 25, pp. 447–457.

Strebulaev, I. (2007), Do Tests of Capital Structure Theory Mean What They Say?, Journal of Finance, 62, pp. 1747-1787.

Welch, I. (2004), Capital Structure and Stock Returns, *Journal of Political Economy*, 112, pp.106-131.

#### Tables and Figures

### Table 1 – Debt decomposition and leverage of the corporate sector – Central Balance Sheet sample

	Debt compo	osition of t of total	he total : debt)	sample (%	Leverage [	Leverage [(Loans + Bonds)/Asset			
	Loans <sup>(1)</sup>	Debt securities	Trade credit <sup>(2)</sup>	Other debt	Total Sample (3)	Average values for the reduced sample <sup>(4)</sup>	Median values for the reduced sample		
1990	50.7	6.5	22.5	19.5	32.1	25.4	3.0		
1991	49.8	5.7	22.4	21.3	30.7	25.2	2.3		
1992	50.3	5.4	21.5	22.2	30.7	25.1	1.8		
1993	55.9	4.9	18.3	20.4	30.1	24.7	0.7		
1994	48.5	7.8	18.6	24.7	26.6	24.5	0.7		
1995	50.8	6.7	23.6	18.4	29.6	22.2	1.6		
1996	49.4	6.7	24.6	18.7	29.5	20.3	2.0		
1997	50.0	8.0	23.4	18.0	29.5	21.0	2.4		
1998	49.2	9.1	23.8	17.4	27.3	18.9	3.7		
1999	50.3	9.9	22.3	16.9	30.4	20.9	4.8		
2000	56.3	4.7	22.1	16.1	31.5	21.0	7.3		
2001	58.0	4.1	21.3	15.7	32.9	22.2	7.2		
2002	59.0	4.1	20.8	15.4	32.7	20.8	7.5		
2003	58.5	3.7	20.6	16.5	32.2	18.9	7.7		
2004	58.3	6.0	19.7	15.0	32.9	19.4	7.1		
2005	55.2	4.4	21.2	18.4	33.1	20.3	5.9		
2006	55.9	4.5	20.6	18.4	34.8	20.7	5.3		
2007	56.5	5.0	19.7	17.9	35.6	19.8	5.2		
Total	55.2	5.4	21.0	17.7	32.5	20.7	4.4		
Number of observations	1 319 202	1 319 202	1 319 202	1 319 202	1 319 202	348 291	348 291		
Number of firm Median number	s 388 356	388 356	388 356	388 356	388 356	48 631	48 631		
of years a firm	3	3	3	3	3	9	9		

Source: Central Balance Sheet database.

Notes: weighted averages except for the last column which reports median values. (1) It includes loans granted by other firms in the same financial group and Accounts Payable to suppliers of fixed assets. (2) It considers only Accounts Payables. (3) It includes loans granted by other firms in the same financial group and Accounts Payables to suppliers of fixed assets. (4) Reduced sample after the application of filters.

Table 2 – Characterization	of data by firm size
----------------------------	----------------------

	Number of	Annual Sales (S) in	Number of	Number of	Leverage
	employees (E )	million euros	observations	firms	(median)
Micro	$E \le 10$	$S \leq 2$	168,188	35,018	0.0
Small	$10 < \mathrm{E} \leq 50$	$2 < S \le 10$	120,734	25,764	8.3
Medium	$50 < \mathrm{E} \leq 250$	$10 < S \le 50$	47,272	9,120	14.2
Large	$\mathrm{E}>250$	$\mathrm{S}>50$	12,097	1,982	12.4

Note: the sum of the column with the number of firms is higher than 50 000 as firms changed from one class size to another.

Age class	Firm age in number of years (Y)	Number of observations	Leverage (median)
1	$Y \leq 7$	$76,\!526$	0.6
2	$7 < Y \le 13$	88,688	4.3
3	$13 < Y \le 22$	92,222	6.2
4	Y > 22	90,855	5.6

Table 3 – Characterization of data by firm age  $% \left( {{{\mathbf{x}}_{i}}} \right)$ 

## Table 4 – Leverage by sector

	Number of	Lever	age
	observations	mean	median
Agriculture	11,647	14.6	5.8
Trade	82,369	12.5	5.4
Construction	$50,\!388$	15.3	6.1
Education	1,303	13.6	4.8
Fishing	1,100	14.1	4.9
Health care	1,709	12.6	2.8
Manufacturing	144,205	12.0	5.3
Mining	3,754	13.2	7.5
Other	2,525	11.1	1.5
Other services	7,032	10.5	1.5
Real estate	3,725	25.1	18.4
Tourism	7,884	12.1	0.0
Transports	$29,\!454$	7.1	-
Utilities	1,196	19.9	9.9
Total	348,291		

	Ν	mean	$\operatorname{sd}$	$\min$	p5	p25	p50	p75	p95	max
Leverage	$348 \ 291$	12.4	16.7	0.0	0.0	0.0	4.4	20.6	47.8	81.1
CF_A	341 670	7.0	12.4	-67.9	-12.0	2.2	6.5	12.6	26.3	51.7
Inv_A	341  400	5.5	9.1	-10.7	0.0	0.0	1.5	6.9	25.5	55.5
Sales growth	276  004	5.9	38.4	-100.0	-46.7	-10.5	2.9	17.6	65.4	293.0
Tang_assets	344 896	26.0	22.6	0.0	0.4	6.9	20.2	40.5	71.5	91.5
ln_Assets	$348 \ 291$	13.2	2.0	0.0	10.2	11.9	13.1	14.5	16.6	23.4
D_group	$348 \ 291$	0.2	0.4	0.0	0.0	0.0	0.0	0.0	1.0	1.0
Liquidity	338  581	55.9	156.3	0.0	0.3	3.1	11.2	38.1	245.8	1851.4
D_RD	$348 \ 291$	0.2	0.4	0.0	0.0	0.0	0.0	0.0	1.0	1.0
Dep_prov_A	$344 \ 938$	6.3	5.9	0.0	0.1	2.0	4.7	8.7	18.2	35.1

Table 5 – Summary statistics

Notes: Leverage is defined as bonds and loans over total assets. CF\_A is net earnings before provisions and depreciation as a % of assets. Inv\_A stands for investment as a % of assets and Tang\_assets is the share of tangible assets in total assets. D\_group is a dummy variable which takes the value one when the firm is consolidated in a group. Liq is a measure of liquidity, defined as short term securities and cash as a % of short term debt. D\_RD is a dummy variable which takes the value one whenever the firm has invested in R&D. Dep prov A is depreciations and provisions for the year as a % of total assets.

Leverage	1									
CF_A	-0.1417*	1								
Inv_A	-0.0135*	$0.2132^{*}$	1							
Sales growth	-0.0215*	$0.2083^{*}$	$0.1390^{*}$	1						
Tang_assets	$0.0494^{*}$	$0.1695^{*}$	$0.4042^{*}$	$0.0471^{*}$	1					
Ln_assets	$0.3247^{*}$	-0.0061*	-0.0142*	$0.0241^{*}$	$0.0585^{*}$	1				
D_group	$0.1226^{*}$	$-0.0189^{*}$	$-0.0247^{*}$	-0.0065*	$0.0404^{*}$	$0.4613^{*}$	1			
Liquidity	-0.1584*	$0.0965^{*}$	-0.0508*	-0.0222*	-0.0672*	-0.1629*	-0.0651*	1		
D_RD	$0.1383^{*}$	$0.0062^{*}$	0.0498*	$0.0077^{*}$	$0.1233^{*}$	$0.3669^{*}$	$0.2454^{*}$	-0.0650*	1	
Dep_prov_A	-0.1045*	0.4019*	$0.3627^{*}$	$0.0614^{*}$	$0.5032^{*}$	-0.1862*	-0.0624*	-0.0047	0.0036	1

#### Table 6 – Correlation matrix

Inv A Sales growth ang asset Ln assets D group Liquidity D RD Dep prov A

Leverage CF A

Notes: \* indicates that the correlation is significant at the 5% level. Leverage is defined as bonds and loans over total assets. CF\_A is net earnings before provisions and depreciation as a % of assets. Inv\_A stands for investment as a % of assets and Tang\_assets is the share of tangible assets in total assets. D\_group is a dummy variable which takes the value one when the firm is consolidated in a group. Liq is a measure of liquidity, defined as short term securities and cash as a % of short term debt. D\_RD is a dummy variable which takes the value one whenever the firm has invested in R&D. Dep\_prov\_A is depreciations and provisions for the year as a % of total assets.

#### Table 7 – Regressions

#### Table 7 - Regressions

Dependent variable: leverage

		All firms	Lagged variables
	CF_A	-0.15	-0.09
	Sales growth	- <i>41.34</i> -0.006	- <i>23.12</i> -0.001
	Tangible assots	- <i>8.17</i> 0.03	- <i>1.24</i> 0.04
		8.09	9.24
	Log assets	4.43 <i>39.45</i>	3.47 27.60
	D_group	-0.16 -1.00	-0.14 -0.82
	Liquidity	-0.004	-0.002
	D_RD	-10.10	-8.00 0.47
	Dep_prov_A	<i>3.39</i> 0.03	<i>3.55</i> -0.005
	Constant	<i>3.10</i> -45.82	-0.45 -33.99
-		-31.58	-20.85
	Number of observations Number of firms	260,794 46,903	200,357 40,979
R-sq:	within between	$0.063 \\ 0.143$	$0.039 \\ 0.129$
	overall	0.120	0.109

Notes: *t*-ratios in italics. Time and firm fixed-effects and robust standard errors are considered. Leverage is defined as bonds and loans over total assets. CF\_A is net earnings before provisions and depreciation as a % of assets. Tang\_assets is the share of tangible assets in total assets. D\_group is a dummy variable which takes the value one when the firm is consolidated in a group. Liq is a measure of liquidity, defined as short term securities and cash as a % of short term debt. D\_RD is a dummy variable which takes the value one whenever the firm has invested in R&D. Dep\_prov\_A is depreciations and provisions for the year as a % of total assets.

### Table 8 – Robustness regressions

	By firm a	size:			By firm	age:			Robustness:						
										All firms	Firms with	Dependent variable:	Dependent variable:	Dependent variable:	Controlling
	Micro	$\mathbf{Small}$	Medium	Large	1st	2nd	3rd	4th	Manufacturi	excluding	positive	dummy	long term	leverage (broad	for debt
	firms	firms	firms	firms	quartile	quartile	quartile	quartile	ng firms	micro	leverage	leverage	leverage	definition)	from group
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)
CF_A	-0.04	-0.12	-0.21	-0.27	-0.03	-0.05	-0.08	-0.12	-0.11	-0.16	-0.16	-0.01	-0.04	-0.09	-0.09
	-9.23	-15.70	-13.56	-8.53	-3.36	-6.91	-9.76	-14.58	-18.19	-23.19	-22.38	-20.52	-14.93	-17.35	-23.20
Sales growth	-0.001	-0.003	-0.004	-0.001	0.000	-0.003	-0.004	0.002	0.000	-0.002	-0.002	0.000	0.000	0.001	-0.001
	-0.48	-2.27	-1.63	-0.18	0.11	-1.91	-2.52	1.08	0.20	-1.94	-1.73	2.39	-0.41	1.11	-1.22
Tangible assets	0.02	0.05	0.04	0.08	0.01	0.04	0.03	0.03	0.04	0.05	0.03	0.01	0.04	0.11	0.04
	3.65	6.44	3.37	3.01	1.38	5.49	3.74	3.85	6.94	8.04	4.60	21.17	13.27	21.03	9.18
Log assets	2.81	3.95	4.47	4.43	1.59	2.49	2.71	4.21	3.70	4.00	3.05	0.52	1.86	4.54	3.49
	15.94	16.90	11.90	6.18	4.92	9.28	10.12	15.43	19.03	21.05	16.12	81.92	21.10	28.43	27.75
D group	-1.05	-0.19	-0.06	-0.02	-0.61	-0.61	0.21	0.10	-0.06	-0.08	-0.15	-0.14	0.12	-0.30	-
	-2.43	-0.69	-0.21	-0.04	-1.18	-1.71	0.69	0.36	-0.24	-0.40	-0.74	-7.18	0.87	-1.55	-
Liquidity	-0.001	-0.004	-0.01	-0.02	-0.002	0.001	-0.001	-0.003	-0.004	-0.005	-0.002	-0.002	0.001	-0.002	-0.002
	-4.58	-6.25	-6.78	-3.25	-2.05	1.28	-2.81	-7.41	-8.53	-9.07	-2.05	-37.36	3.60	-7.04	-8.61
D RD	0.36	0.45	0.48	0.52	-0.07	-0.17	0.35	0.80	0.61	0.52	0.31	0.13	0.25	0.51	0.46
	1.19	2.29	2.04	1.15	-0.17	-0.69	1.56	3.58	3.47	3.55	2.07	7.45	2.40	3.53	3.50
Dep prov A	0.04	-0.01	-0.03	-0.04	0.03	-0.02	0.01	-0.03	-0.02	-0.03	0.04	-0.01	-0.03	-0.02	0.00
	2.85	-0.66	-1.02	-0.53	1.21	-1.33	0.61	-1.31	-1.42	-1.51	2.52	-6.17	-4.00	-1.13	-0.41
Debt from group	-	-	-	-	-	-	-	-	-	-	-	-	-	-	0.00
	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-3.23
Constant	-25.03	-40.85	-51.99	-54.47	-9.49	-20.67	-23.95	-44.28	-38.09	-41.95	-21.69	-6.23	-21.82	-48.45	-34.24
	-11.98	-13.30	-8.80	-4.55	-2.38	-5.64	-6.88	-11.90	-14.18	-15.93	-8.55	-73.51	-18.95	-23.08	-20.99
Namel an af															
Number of	09 570	74.004	20.000	0.100	00.449	F 4 0 1 0	F0 F70	50.994	01 740	110 770	100.004	000 057	000 957	000 957	000 057
observations	83,578	74,084	32,926	9,169	28,443	54,018	58,572	59,324	91,740	116,779	126,064	200,357	200,357	200,357	200,357
Number of firms	23,532	17,687	6,610	1,571	13,306	19,118	19,614	13,080	16,115	22,750	30,458	40,979	40,979	40,979	40,979
within	0.024	0.052	0.065	0.071	0.011	0.020	0.023	0.042	0.045	0.054	0.034	-	0.022	0.067	0.039
between	0.127	0.121	0.093	0.070	0.096	0.109	0.121	0.156	0.159	0.093	0.032	-	0.092	0.117	0.131
	0.100	0.100	0.000	0.050	o oo <b>-</b>										

Notes: t-ratios in italics. Time and firm fixed-effects and robust standard errors are considered. All explanatory variables are lagged by 1 year. Leverage is defined as bonds and loans over total assets. CF\_A is net earnings before provisions and depreciation as a % of assets. Tang\_assets is the share of tangible assets in total assets. D\_group is a dummy variable which takes the value one when the firm is consolidated in a group. Liq is a measure of liquidity, defined as short term securities and cash as a % of short term debt. D\_RD is a dummy variable which takes the value one whenever the firm has invested in R&D. Dep\_prov\_A is depreciations and provisions for the year as a % of total assets. The dummy leverage takes the value one when firms have positive leverage. Long term leverage considers long term loans and bonds as a percentage of total assets. The broad definition of leverage includes debt from suppliers of tangible assets. Debt from group considers the total amount of debt obtained within the economic group a firm belongs to.

### Table 9 – Target leverage ratio – two step regressions

		Depende	nt variab.	le: chang	je in levera	age			<b>m</b>							
			<b>D</b> (1	•					Two-st	ep regres	sions					
		Full sample	By firm Micro firms	size: Small firms	Medium firms	Large firms	By firm 1st quartile	age: 2nd quartile	3rd quartile	4th quartile	Kobustnes Manufac turing firms	All firms excluding micro	Full sample (dummy leverage)	Firms with positive leverage	Dependent variable: long term leverage	Dependent variable: leverage (broad definition)
		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)
	Adjustment	0.62 <i>131.81</i>	0.73 <i>90.57</i>	0.62 <i>85.75</i>	0.54 54.08	0.45 <i>23.59</i>	1.00 78.77	0.86 <i>90.14</i>	0.75 <i>85.00</i>	0.56 <i>65.67</i>	0.59 <i>87.43</i>	0.56 <i>101.28</i>	0.60 <i>128.09</i>	0.58 115.87	0.79 <i>140.66</i>	0.59 <i>129.19</i>
	D.CF_A t	-0.090 - <i>33.85</i>	-0.052 - <i>16.20</i>	-0.120 - <i>23.16</i>	-0.188 - <i>18.40</i>	-0.234 - <i>12.22</i>	-0.064 - <i>10.75</i>	-0.080 - <i>16.42</i>	-0.089 - <i>16.11</i>	-0.097 - <i>17.61</i>	-0.092 - <i>25.53</i>	-0.144 - <i>32.58</i>	-0.089 - <i>33.62</i>	-0.168 - <i>33.12</i>	-0.014 - <i>10.77</i>	-0.095 - <i>32.65</i>
	D.CF_A <sub>t-1</sub>	-0.030 - <i>13.34</i>	-0.010 - <i>3.81</i>	-0.041 <i>-8.72</i>	-0.079 <i>-9.25</i>	-0.094 - <i>6.01</i>	-0.025 -4.27	-0.029 -6.58	-0.030 - <i>6.44</i>	-0.042 -9.57	-0.034 - <i>10.61</i>	-0.056 - <i>14.03</i>	-0.029 - <i>12.79</i>	-0.068 - <i>15.38</i>	-0.010 -8.47	-0.040 -15.16
	Inv_A t	0.036 <i>11.00</i>	0.039 <i>8.41</i>	0.026 4.75	0.043 <i>4.85</i>	0.066 <i>2.97</i>	0.028 <i>3.31</i>	0.018 <i>2.94</i>	0.027 <i>4.07</i>	0.047 <i>7.00</i>	0.038 <i>8.51</i>	0.032 <i>7.01</i>	0.036 <i>11.05</i>	0.049 <i>9.61</i>	0.011 5.69	0.192 46.29
	Inv_A <sub>t-1</sub>	0.022 <i>7.76</i>	0.013 <i>3.06</i>	0.022 <i>4.92</i>	0.050 <i>6.47</i>	0.081 <i>3.99</i>	0.003 <i>0.44</i>	0.020 <i>3.57</i>	0.026 <i>4.27</i>	0.020 <i>3.69</i>	0.022 5.87	0.031 <i>7.94</i>	0.020 <i>7.18</i>	0.029 <i>6.83</i>	0.010 5.40	0.029 <i>8.53</i>
	Dummy leverage	-	-	-	-	- -	-	-	-	-	-	-	2.391 <i>24.90</i>	-	-	-
	Constant	-0.84 -5.47	0.27 <i>0.91</i>	-1.22 - <i>5.32</i>	-2.25 - <i>8.99</i>	-3.18 -4.31	0.82 1.11	0.07 <i>0.28</i>	-0.75 - <i>2.18</i>	-1.23 - <i>4.35</i>	-1.02 - <i>9.00</i>	-1.45 -8.13	-2.18 - <i>13.37</i>	0.51 2.15	-0.69 - <i>6.84</i>	-2.02 - <i>12.07</i>
	Number of observations Number of firms	190,512 39,861	77,362 22,436	72,177 17,319	32,046 6,509	8,927 1,549	26,406 12,567	51,290 18,422	56,295 19,104	56,521 12,706	88,026 15,774	113,150 22,376	190,512 39,861	121,418 29,741	197,708 40,825	194,351 40,421
R-sq:	within between	0.36 0.07	0.43 0.10	0.37	0.33 0.05	0.28 0.05	0.58 0.12	0.50 0.10	0.43 0.08	0.33	0.35 0.07	0.33 0.06	0.37 0.08	0.33 0.07	0.56 0.27	0.37 0.07
	overan	0.16	0.20	0.15	0.12	0.11	0.20	0.18	0.14	0.12	0.16	0.14	0.17	0.13	0.39	0.17

Dependent variable: change in leverage

Notes: t-ratios in italics. Time and firm fixed-effects and robust standard errors are considered. Leverage is defined as bonds and loans over total assets. CF\_A is net earnings before provisions and depreciation as a % of assets. Inv\_A stands for investment as a % of assets. All other variables defined in previous tables. For columns (14) and (15), specific target leverage ratios were considered (i.e., for long term leverage and for the broad definition of leverage, respectively).

Table $10 - Positive versus n$	negative	adjustments
--------------------------------	----------	-------------

#### Panel A

	%	of observat	ions	Le	verage (med	ian)	Ta	arget (medi	an)	Adj	Adjustment (median)			
Firm size	Negative adjustment	Positive adjustment with leverage	Positive adjustment without leverage											
M:	20.6	99 F	20.0	92.0	7 7	0.0	16.9	16.6	0.5	10.9	7.0	95		
Micro firms	32.0	28.5	38.8	23.9	(.(	0.0	10.3	10.0	2.5	-12.3	7.8	2.5		
Small firms	33.0	37.9	28.6	28.6	9.3	0.0	19.5	19.0	4.6	-10.9	8.9	4.6		
Medium firms	35.1	44.5	20.4	31.9	11.7	0.0	22.8	22.2	8.1	-10.1	9.8	8.1		
Large firms	28.7	50.6	20.7	35.8	10.2	0.0	26.3	25.9	11.9	-10.8	13.5	11.9		
Total	33.3	36.3	30.4	28.5	9.4	0.0	19.1	19.0	4.3	-11.1	9.0	4.3		

Panel B

	% of observations			Leverage (median)		Target (median)			Adjustment (median)			
	Negative adjustment	Positive adjustment with	Positive adjustment without									
Sector		leverage	leverage									
Agriculture Trade	39.0 33.1	32.0 35.0	$29.1 \\ 31.9$	31.6 28.7	$8.6 \\ 9.3$	$0.0 \\ 0.0$	$18.6 \\ 19.0$	18.8 19.0	$4.4 \\ 4.4$	-14.7 -11.3	$8.8 \\ 8.9$	$4.4 \\ 4.4$
Construction	38.3	34.2	27.5	30.6	10.3	0.0	18.9	18.6	4.0	-14.4	8.8	4.0
Education	33.3	38.3	28.5	32.2	7.5	0.0	19.7	19.3	5.0	-12.6	9.7	5.0
Fishing	34.7	37.0	28.3	31.2	9.2	0.0	20.2	21.8	5.4	-15.3	11.2	5.4
Health care	31.6	36.1	32.3	28.3	8.0	0.0	19.1	18.9	6.3	-12.7	9.0	6.3
Manufacturing	g 33.0	38.3	28.7	27.5	9.6	0.0	19.3	19.1	4.1	-9.9	8.9	4.1
Mining	32.8	41.2	26.0	27.7	9.2	0.0	19.9	19.9	5.4	-10.6	10.0	5.4
Other	26.5	35.9	37.6	26.3	9.9	0.0	19.3	21.7	5.7	-10.0	9.9	5.7
Other services	24.5	37.9	37.5	28.1	6.4	0.0	19.5	19.9	6.5	-11.4	11.2	6.5
Real estate	52.0	26.8	21.2	43.9	9.5	0.0	23.0	22.4	8.3	-22.1	9.9	8.3
Tourism	30.4	33.4	36.2	33.3	7.5	0.0	20.1	21.1	4.9	-16.4	11.2	4.9
Transports	24.0	34.5	41.5	23.9	7.4	0.0	17.0	18.2	3.7	-10.3	9.3	3.7
Utilities	43.8	29.8	26.4	40.6	8.0	0.0	25.7	25.0	8.0	-17.1	11.9	8.0
	33.3	36.3	30.4	28.5	9.4	0.0	19.1	19.0	4.3	-11.1	9.0	4.3

Notes: A negative adjustment means that the firm's current leverage ratio is above its target. Firms that do not make any adjustments from one period to another are excluded from this analysis. Observations labeled as with or without leverage are classified using the lagged values for this variable.

	_	Dependent variable: change in leverage					
		Two-step regressions					
		×					
		Negative adjustment	Positive adjustment with leverage	Positive adjustment without leverage			
		(1)	(2)	(3)			
	Adjustment	0.63 68.45	0.53 68.03	0.73 62.66			
	D.CF_A $_{\rm t}$	-0.202	-0.125	-0.035			
		-23.65	-21.86	-10.78			
	$D.CF_A_{t-1}$	-0.081	-0.053	0.012			
		-10.74	-10.43	3.94			
	Inv_A <sub>t</sub>	-0.052	0.078	0.050			
		-5.90	12.55	9.63			
	Inv A <sub>t-1</sub>	0.053	0.029	-0.001			
		7.11	5.66	-0.35			
	Constant	5.96	-4.79	-1.75			
		13.26	-17.64	-9.33			
	Number of observations	57,026	$63,\!124$	$51,\!970$			
	Number of firms	19,734	$20,\!538$	18,117			
D	: (1) :	0.96	0.15	0.24			
к-sq:	within	0.20	0.15	0.34			
	between	0.06	0.01	0.23			
	overall	0.10	0.04	0.25			

### Table 11 - Target leverage ratio with negative and positive adjustments

Notes: t-ratios in italics. A negative adjustment means that the firm's current leverage ratio is above its target. Time and firm fixed-effects and robust standard errors are considered. Leverage is defined as bonds and loans over total assets. All other variables defined in previous tables. We consider that the firm has to make a negative adjustment (column 1) if the target leverage ratio is below the observed leverage ratio and the opposite for positive adjustments. There are two different types of positive adjustments: for firms with and without debt in the previous period (columns 2 and 3, respectively).

	Spe	Speed of adjustment			
Firm size	Negative adjustment	Negative adjustment with leverage			
Micro firms	0.72	0.60	0.86		
Small firms	0.60	0.57	0.68		
Medium firms	0.58	0.53	0.58		
Large firms	0.54	0.46	0.43		
Total	0.63	0.53	0.73		

### Table 12 – Positive versus negative adjustments by firm size

#### Panel B

Panel A

	Speed of adjustment (for adjustment above the median)				
Firm size	Negative adjustment	Positive adjustment with leverage	Positive adjustment without leverage		
Micro firms	0.81	0.73	0.80		
Small firms	0.68	0.69	0.62		
Medium firms	0.65	0.57	0.59		
Large firms	0.72	0.48	0.36		
Total	0.72	0.56	0.61		

Notes: the speed of adjustment is the coefficient for the variable adjustment in each regression (each cell refers to a different regression). All these coefficients are significant at a 1% confidence level. A negative adjustment means that the firm's current leverage ratio is above its target.

	Dependent variable: time to target					
	Full sample	Negative adjustment	Positive adjustment with leverage	Positive adjustment without leverage		
	(1)	(2)	(3)	(4)		
CF_A t	0.999	1.022	0.990	0.997		
Sales growth $_{\rm t-1}$	1.001 2.17	1.001 2.42	1.000 0.64	0.999		
Tangible assets $_{\rm t}$	1.003 5.75	1.002 2.19	1.001 0.97	1.004 2.39		
Log assets $_{\rm t}$	1.148 <i>18.97</i>	1.055 <i>4.13</i>	0.982 -1.46	1.061 <i>2.35</i>		
D_group $_{\rm t}$	0.861 -4.65	1.047 <i>0.93</i>	0.913 - <i>1.91</i>	0.662 - <i>3.45</i>		
Liquidity $_{\rm t}$	0.995 - <i>19.35</i>	0.999 - <i>3.96</i>	0.997 - <i>7.35</i>	0.995 - <i>8.32</i>		
D_RD <sub>t</sub>	1.159 <i>5.14</i>	1.060 1.28	1.138 <i>3.08</i>	1.271 2.46		
Dep_prov_A t	1.009 <i>3.38</i>	1.000 <i>0.07</i>	1.014 <i>3.39</i>	1.013 <i>1.63</i>		
Log pseudo likelihood	-73147	-24236	-28052	-6594		
Number of observations	225753	45262	46399	49536		
Number of failures Number of subjects	46833 7837	18301 3001	18974 3470	17673 835		
Time at risk	226854	45262	46399	49536		

#### Table 13 – Cox proportional hazard regressions

Notes: z-scores in italics. The dependent variable is the time a firm takes to reach its target leverage ratio, considering that a firm is close to the target when the observed leverage ratio is within the interval [90%,110%] of the initial estimated target. An estimated coefficient lower than 1 should be interpreted as contributing a longer time until the firm reaches its target. All variables defined in previous tables. We consider that the firm has to make a negative adjustment (column 2) if the target leverage ratio is below the observed leverage ratio and the opposite for positive adjustments. There are two different types of positive adjustments: for firms with and without debt in the previous year (columns 3 and 4, respectively).

### **WORKING PAPERS**

#### 2010

- 1/10 MEASURING COMOVEMENT IN THE TIME-FREQUENCY SPACE — António Rua
- 2/10 EXPORTS, IMPORTS AND WAGES: EVIDENCE FROM MATCHED FIRM-WORKER-PRODUCT PANELS — Pedro S. Martins, Luca David Opromolla
- 3/10 NONSTATIONARY EXTREMES AND THE US BUSINESS CYCLE — Miguel de Carvalho, K. Feridun Turkman, António Rua
- 4/10 EXPECTATIONS-DRIVEN CYCLES IN THE HOUSING MARKET — Luisa Lambertini, Caterina Mendicino, Maria Teresa Punzi
- 5/10 COUNTERFACTUAL ANALYSIS OF BANK MERGERS — Pedro P. Barros, Diana Bonfim, Moshe Kim, Nuno C. Martins
- 6/10 THE EAGLE. A MODEL FOR POLICY ANALYSIS OF MACROECONOMIC INTERDEPENDENCE IN THE EURO AREA — S. Gomes, P. Jacquinot, M. Pisani
- 7/10 A WAVELET APPROACH FOR FACTOR-AUGMENTED FORECASTING — António Rua
- 8/10 EXTREMAL DEPENDENCE IN INTERNATIONAL OUTPUT GROWTH: TALES FROM THE TAILS — Miguel de Carvalho, António Rua
- 9/10 TRACKING THE US BUSINESS CYCLE WITH A SINGULAR SPECTRUM ANALYSIS — Miguel de Carvalho, Paulo C. Rodrigues, António Rua
- 10/10 A MULTIPLE CRITERIA FRAMEWORK TO EVALUATE BANK BRANCH POTENTIAL ATTRACTIVENESS — Fernando A. F. Ferreira, Ronald W. Spahr, Sérgio P. Santos, Paulo M. M. Rodrigues
- 11/10 THE EFFECTS OF ADDITIVE OUTLIERS AND MEASUREMENT ERRORS WHEN TESTING FOR STRUCTURAL BREAKS IN VARIANCE
  - Paulo M. M. Rodrigues, Antonio Rubia
- 12/10 CALENDAR EFFECTS IN DAILY ATM WITHDRAWALS — Paulo Soares Esteves, Paulo M. M. Rodrigues
- 13/10 MARGINAL DISTRIBUTIONS OF RANDOM VECTORS GENERATED BY AFFINE TRANSFORMATIONS OF INDEPENDENT TWO-PIECE NORMAL VARIABLES
  - Maximiano Pinheiro
- 14/10 MONETARY POLICY EFFECTS: EVIDENCE FROM THE PORTUGUESE FLOW OF FUNDS — Isabel Marques Gameiro, João Sousa
- 15/10 SHORT AND LONG INTEREST RATE TARGETS
  - Bernardino Adão, Isabel Correia, Pedro Teles
- 16/10 FISCAL STIMULUS IN A SMALL EURO AREA ECONOMY — Vanda Almeida, Gabriela Castro, Ricardo Mourinho Félix, José Francisco Maria
- 17/10 FISCAL INSTITUTIONS AND PUBLIC SPENDING VOLATILITY IN EUROPE
  - Bruno Albuquerque

18/10 GLOBAL POLICY AT THE ZERO LOWER BOUND IN A LARGE-SCALE DSGE MODEL

- S. Gomes, P. Jacquinot, R. Mestre, J. Sousa
- 19/10 LABOR IMMOBILITY AND THE TRANSMISSION MECHANISM OF MONETARY POLICY IN A MONETARY UNION — Bernardino Adão, Isabel Correia
- 20/10 TAXATION AND GLOBALIZATION
  - Isabel Correia
- 21/10 TIME-VARYING FISCAL POLICY IN THE U.S.
  - Manuel Coutinho Pereira, Artur Silva Lopes
- 22/10 DETERMINANTS OF SOVEREIGN BOND YIELD SPREADS IN THE EURO AREA IN THE CONTEXT OF THE ECONOMIC AND FINANCIAL CRISIS
  - Luciana Barbosa, Sónia Costa
- 23/10 FISCAL STIMULUS AND EXIT STRATEGIES IN A SMALL EURO AREA ECONOMY — Vanda Almeida, Gabriela Castro, Ricardo Mourinho Félix, José Francisco Maria
- 24/10 FORECASTING INFLATION (AND THE BUSINESS CYCLE?) WITH MONETARY AGGREGATES — João Valle e Azevedo, Ana Pereira
- 25/10 THE SOURCES OF WAGE VARIATION: AN ANALYSIS USING MATCHED EMPLOYER-EMPLOYEE DATA — Sónia Torres, Pedro Portugal, John T.Addison, Paulo Guimarães
- 26/10 THE RESERVATION WAGE UNEMPLOYMENT DURATION NEXUS — John T. Addison, José A. F. Machado, Pedro Portugal
- 27/10 BORROWING PATTERNS, BANKRUPTCY AND VOLUNTARY LIQUIDATION — José Mata, António Antunes, Pedro Portugal
- 28/10 THE INSTABILITY OF JOINT VENTURES: LEARNING FROM OTHERS OR LEARNING TO WORK WITH OTHERS — José Mata, Pedro Portugal
- 29/10 THE HIDDEN SIDE OF TEMPORARY EMPLOYMENT: FIXED-TERM CONTRACTS AS A SCREENING DEVICE — Pedro Portugal, José Varejão
- 30/10 TESTING FOR PERSISTENCE CHANGE IN FRACTIONALLY INTEGRATED MODELS: AN APPLICATION TO WORLD INFLATION RATES
  - Luis F. Martins, Paulo M. M. Rodrigues
- 31/10 EMPLOYMENT AND WAGES OF IMMIGRANTS IN PORTUGAL
  - Sónia Cabral, Cláudia Duarte
- 32/10 EVALUATING THE STRENGTH OF IDENTIFICATION IN DSGE MODELS. AN A PRIORI APPROACH — Nikolay Iskrev
- 33/10 JOBLESSNESS — José A. F. Machado, Pedro Portugal, Pedro S. Raposo

### 2011

- 1/11 WHAT HAPPENS AFTER DEFAULT? STYLIZED FACTS ON ACCESS TO CREDIT — Diana Bonfim, Daniel A. Dias, Christine Richmond
- 2/11 IS THE WORLD SPINNING FASTER? ASSESSING THE DYNAMICS OF EXPORT SPECIALIZATION — João Amador

3/11 UNCONVENTIONAL FISCAL POLICY AT THE ZERO BOUND Isabel Correia, Emmanuel Farhi, Juan Pablo Nicolini, Pedro Teles 4/11 MANAGERS' MOBILITY, TRADE STATUS, AND WAGES — Giordano Mion, Luca David Opromolla 5/11 FISCAL CONSOLIDATION IN A SMALL EURO AREA ECONOMY — Vanda Almeida, Gabriela Castro, Ricardo Mourinho Félix, José Francisco Maria 6/11 CHOOSING BETWEEN TIME AND STATE DEPENDENCE: MICRO EVIDENCE ON FIRMS' PRICE-REVIEWING STRATEGIES — Daniel A. Dias, Carlos Robalo Margues, Fernando Martins 7/11 WHY ARE SOME PRICES STICKIER THAN OTHERS? FIRM-DATA EVIDENCE ON PRICE ADJUSTMENT LAGS — Daniel A. Dias, Carlos Robalo Margues, Fernando Martins, J. M. C. Santos Silva 8/11 LEANING AGAINST BOOM-BUST CYCLES IN CREDIT AND HOUSING PRICES — Luisa Lambertini, Caterina Mendicino, Maria Teresa Punzi 9/11 PRICE AND WAGE SETTING IN PORTUGAL LEARNING BY ASKING — Fernando Martins 10/11 ENERGY CONTENT IN MANUFACTURING EXPORTS: A CROSS-COUNTRY ANALYSIS — João Amador 11/11 ASSESSING MONETARY POLICY IN THE EURO AREA: A FACTOR-AUGMENTED VAR APPROACH - Rita Soares 12/11 DETERMINANTS OF THE EONIA SPREAD AND THE FINANCIAL CRISIS - Carla Soares, Paulo M. M. Rodrigues 13/11 STRUCTURAL REFORMS AND MACROECONOMIC PERFORMANCE IN THE EURO AREA COUNTRIES: A MODEL-BASED ASSESSMENT - S. Gomes, P. Jacquinot, M. Mohr, M. Pisani 14/11 RATIONAL VS. PROFESSIONAL FORECASTS — João Valle e Azevedo, João Tovar Jalles 15/11 ON THE AMPLIFICATION ROLE OF COLLATERAL CONSTRAINTS — Caterina Mendicino 16/11 MOMENT CONDITIONS MODEL AVERAGING WITH AN APPLICATION TO A FORWARD-LOOKING MONETARY POLICY REACTION FUNCTION — Luis F. Martins 17/11 BANKS' CORPORATE CONTROL AND RELATIONSHIP LENDING: EVIDENCE FROM RETAIL LOANS — Paula Antão, Miguel A. Ferreira, Ana Lacerda 18/11 MONEY IS AN EXPERIENCE GOOD: COMPETITION AND TRUST IN THE PRIVATE PROVISION OF MONEY — Ramon Marimon, Juan Pablo Nicolini, Pedro Teles 19/11 ASSET RETURNS UNDER MODEL UNCERTAINTY: EVIDENCE FROM THE EURO AREA, THE U.K. AND THE U.S. — João Sousa, Ricardo M. Sousa 20/11 INTERNATIONAL ORGANISATIONS' VS. PRIVATE ANALYSTS' FORECASTS: AN EVALUATION — Ildeberta Abreu 21/11 HOUSING MARKET DYNAMICS: ANY NEWS? — Sandra Gomes, Caterina Mendicino

- 22/11 MONEY GROWTH AND INFLATION IN THE EURO AREA: A TIME-FREQUENCY VIEW — António Rua
- 23/11 WHY EX(IM)PORTERS PAY MORE: EVIDENCE FROM MATCHED FIRM-WORKER PANELS — Pedro S. Martins, Luca David Opromolla
- 24/11 THE IMPACT OF PERSISTENT CYCLES ON ZERO FREQUENCY UNIT ROOT TESTS — Tomás del Barrio Castro, Paulo M.M. Rodrigues, A.M. Robert Taylor
- 25/11 THE TIP OF THE ICEBERG: A QUANTITATIVE FRAMEWORK FOR ESTIMATING TRADE COSTS — Alfonso Irarrazabal, Andreas Moxnes, Luca David Opromolla
- 26/11 A CLASS OF ROBUST TESTS IN AUGMENTED PREDICTIVE REGRESSIONS — Paulo M.M. Rodrigues, Antonio Rubia
- 27/11 THE PRICE ELASTICITY OF EXTERNAL DEMAND: HOW DOES PORTUGAL COMPARE WITH OTHER EURO AREA COUNTRIES?
  - Sónia Cabral, Cristina Manteu
- 28/11 MODELING AND FORECASTING INTERVAL TIME SERIES WITH THRESHOLD MODELS: AN APPLICATION TO S&P500 INDEX RETURNS
  - Paulo M. M. Rodrigues, Nazarii Salish
- 29/11 DIRECT VS BOTTOM-UP APPROACH WHEN FORECASTING GDP: RECONCILING LITERATURE RESULTS WITH INSTITUTIONAL PRACTICE
  - Paulo Soares Esteves
- 30/11 A MARKET-BASED APPROACH TO SECTOR RISK DETERMINANTS AND TRANSMISSION IN THE EURO AREA *— Martín Saldías*
- 31/11 EVALUATING RETAIL BANKING QUALITY SERVICE AND CONVENIENCE WITH MCDA TECHNIQUES: A CASE STUDY AT THE BANK BRANCH LEVEL
  - Fernando A. F. Ferreira, Sérgio P. Santos, Paulo M. M. Rodrigues, Ronald W. Spahr

### 2012

- 1/12 PUBLIC-PRIVATE WAGE GAPS IN THE PERIOD PRIOR TO THE ADOPTION OF THE EURO: AN APPLICATION BASED ON LONGITUDINAL DATA
   Maria Manuel Campos, Mário Centeno
- 2/12 ASSET PRICING WITH A BANK RISK FACTOR — João Pedro Pereira, António Rua
- 3/12 A WAVELET-BASED ASSESSMENT OF MARKET RISK: THE EMERGING MARKETS CASE — António Rua, Luis C. Nunes
- 4/12 COHESION WITHIN THE EURO AREA AND THE U. S.: A WAVELET-BASED VIEW — António Rua, Artur Silva Lopes
- 5/12 EXCESS WORKER TURNOVER AND FIXED-TERM CONTRACTS: CAUSAL EVIDENCE IN A TWO-TIER SYSTEM — Mário Centeno, Álvaro A. Novo
- 6/12 THE DYNAMICS OF CAPITAL STRUCTURE DECISIONS — Paula Antão, Diana Bonfim